Author's response to reviews

Title: Predictors of Pain and Use of Pain medications following Primary Total Hip Arthroplasty (THA): 5,707 THAs at 2-years and 3,289 THAs at 5-years

Authors:

Jasvinder A Singh (jasvinder.md@gmail.com)
David G Lewallen (david.lewallen@mayo.edu)

Version: 3 Date: 1 May 2010

Author's response to reviews:

Here are our point-by-point responses to Editor’s suggestions:

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is “First name Initial Last name”? (e.g. Joe F Bloggs). Please also remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submission system.
A: Changed as suggested

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.
A: Changed as suggested

Financial Conflict statement: Please rename ’Competing Interests’ and move to the end of the manuscript, between the conclusions and authors’ contributions.
A: Changed as suggested

Abstract: please ensure that the Abstract in the manuscript matches exactly what is in the submission system.
A: Done as suggested

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
A: Done as suggested

Tables: tables should fit into an A4 page (maximum size 176mm x 230mm), and the orientation must be portrait only. Tables of up to 3 pages are allowed.
If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Authors? information (optional); Acknowledgements (optional); References

---

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page.

Tracking: Please remove all tracked changes and comments from the manuscript, including strikethrough, underlined, highlighted and coloured text, ensuring that only the final version of all text remains. This may involve accepting or rejecting the tracked changes.

Title page: Please remove running title, Keywords and word counts.

Email address: please include email addresses for all the authors on the title page, using the following format:

AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Additional file 1: This is not suitable for the additional files section, please
A: We have removed this file as suggested.

References: Please check if references 25 and 26 'In Press' have now been published. If so, please provide full citation details. You can check here:
A: Done as suggested

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
A: Done as suggested

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
A: Done as suggested

-----------------------------